Author's response to reviews

Title: Blind Loop: Rare but Important Surgical Complication

Authors:

  John Michael DiBianco (John.DiBianco@yahoo.com)
  James A Palesty (alexander.palesty@stmh.org)
  Robert A Brenes (brenes3@yahoo.com)

Version: 8  Date: 22 July 2014

Author's response to reviews: see over
Dear Mr. Nonato,

Thank you very much for your comments and advice. I have detailed the changes made below as per your suggestions:

1. I have removed the author information form the Title Page and placed if after the Authors Contributions section. I have moved the Keywords section from the title page to after the Abstract.
2. I have moved the Abstract from the title page to the next page, before the Keywords and the Manuscript.
3. I have named the first paragraph of the manuscript with Background.
   a. I have also considered the Review and Discussion sections as subdivisions of the case report. Please let me know if I should change their format to indicate them as such.
4. I have placed the figure at end of the document In the figures section.

Thank you for your consideration!

Sincerely,

- John Michael DiBianco